Author's response to reviews

**Title:** The Hall Technique; a randomised control clinical trial of a novel method of managing carious primary molars in general dental practice: outcomes at two years.

**Authors:**
- Nicola P T Innes (n.p.innes@dundee.ac.uk)
- Dafydd J P Evans (d.j.p.evans@dundee.ac.uk)
- David R Stirrups (d.r.stirrups@dundee.ac.uk)

**Version:** 2 **Date:** 5 July 2007

**Author's response to reviews:**

Clinical trial number has been included as requested.